

PEER REVIEW HISTORY

Gynecology and Obstetrics Clinical Medicine publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

ARTICLE DETAILS

ARTICLE DETAILS	
TITLE (PROVISIONAL)	MINIMALLY INVASIVE METHODS WITH NO FIBROID SURGICAL
	REMOVAL TO TREAT UTERINE FIBROIDS: UTILITY OR ADVERTISING?
AUTHORS	Tinelli, Andrea
	VERSION 1 - REVIEW
REVIEWER	
REVIEW RETURNED	
GENERAL COMMENTS	
REVIEWER	
REVIEW RETURNED	
GENERAL COMMENTS	
VERSION 1 – AUTHOR RESPONSE	
VERSION 2 – REVIEW	
REVIEWER	
REVIEW RETURNED	
GENERAL COMMENTS	
REVIEWER	
REVIEW RETURNED	
GENERAL COMMENTS	

VERSION 2 – AUTHOR RESPONSE

VERSION 3 – REVIEW

REVIEWER	
REVIEW RETURNED	
GENERAL COMMENTS	
REVIEWER	
REVIEW RETURNED	
GENERAL COMMENTS	

VERSION 3 – AUTHOR RESPONSE